Author's response to reviews

Title: Cigarette, cigar and pipe smoking, passive smoke exposure, and risk of pancreatic cancer: a population-based study in the San Francisco Bay Area

Authors:

Gregory J Tranah (gtranah@psg.ucsf.edu)
Elizabeth A Holly (elizabeth.holly@ucsf.edu)
Furong Wang (furong.wang@ucsf.edu)
Paige M Bracci (paige.bracci@ucsf.edu)

Version: 5 Date: 4 April 2011

Author's response to reviews:

The following changes or updates have been completed:

Major revisions (we require the author to make these changes)

Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the country in the affiliation details.

Title page: Please remove the text 'Word count: Abstract 202, Text 2700' from the title page.

Funding: Please move the statement of funding to the Acknowledgements section and remove it from the title page.

Abstract: Please rename the section titled 'Objectives' in the abstract 'Background'.

Introduction: Please rename this section 'Background'.

Materials and methods: Please rename this section 'Methods'.

Structure: Please insert a heading titled 'Conclusions' after the discussion.

References: please provide the full citation details for references 7, 12 and 31. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.